Author's response to reviews

Title: Psychological distress among Bam earthquake survivors in Iran: a population-based study

Authors:

Ali Montazeri (ali@jdcord.jd.ac.ir)
Hamid Baradaran (baradaran98@yahoo.com)
Sepideh Omidvari (somidvari@ihsr.ac.ir)
Seyed Ali Azin (aazin@ihsr.ac.ir)
Mehdi Ebadi (mebadi@ihsr.ac.ir)
Gholamreza Garmaroudi (ggarmaroudi@ihsr.ac.ir)
Amir Mahmood Harirchi (hariram@hbi.or.ir)
Mohammad Shariati (mshariati@ihsr.ac.ir)

Version: 6 Date: 20 December 2004

Author's response to reviews:

Jane Borthwick PhD
Assistant Editor
BMC Journals

Dear Dr. Borthwick,

Re: MS: 3489021124253959
Psychological distress among Bam earthquake survivors in Iran: a population-based study

Thank you for your kind e-mail. The formatting changes have been made to the above manuscript as requested. Hope you find it satisfactory.

Wish you all the best.
Regards
Ali Montazeri